Notes of the RICH1 Phone meeting 13/05/2005
People Present:

Bristol: Richard

CERN:  Nick, Werner, Carmelo (part time)
IC: Dave, Trevor 

Oxford: Neville, Rohan
RAL, Milano: excused

Apologies for absence:

CERN: Fabio
IC: Bill, Geoff

Agenda/Notes on CDS
The RICH1 phone-in meeting agenda and notes are on CDS.

1. Magnetic Shielding: 
1.1 Status:

All components required for the installation are at CERN. The temporary concrete bridge will be installed 13 May. Last-minute drilling mods to the shielding have been completed. These include (i) repositioning of upper box support holes to compensate for wrong z-coord of concrete wall (ii) counterboring of the proud bolts on the shelf. The magnetic shield components are bolted together and are placed at the top of the LHCb access shaft. They are ready to be moved into the pit. Work schedule has been delayed. New date for completion of installation by 27 May. 
1.2 Magnetic Field Measurements:
The magnetic field measurements are probably delayed by one further week: start ~4th June. The B-machine will be available until end June and then will be given to ALICE.
1.3 Equipment to remove side plates:

Rolf will call a meeting to discuss this, where we should consider other requirements, such as handling the HPD assemblies (500kg).  Bill will prepare requirements (action Bill)
 2. Gas Enclosure:
Following meeting with Imperial and RAL on 5th May, it was decided to increase the length by 12mm (to z=2137) of the GE to allow direct mounting of the exit window. Rohan has included this in his drawings for material cutting. No news whether or not NAB will increase cost for extended GE. All paperwork is ready for the order is to be sent out  - just waiting for new quote. (action Tony/Rohan – get new quote, deliver cutting drawings to NAB)

Remaining work to do on the gas enclosure: holes for aerogel, plane mirrors, and beampipe handling. Meetings planned at CERN on 17 May, to finalize these. Trevor to hand over hole positions to Rohan on 20May, Rohan to complete drawings by 27May, when they will be posted to EDMS for approval. (action Trevor, Rohan)
3. VELO Seal
Dave reported that Mike confirmed the timetable for the seal-to-velo manufacture was compatible with a trial assembly to the GE prior to installation in the cavern. 
Dave has alerted the VELO team of Mike’s drawings on EDMS – noted an inconsistency with the VELO exit window clamping ring outer diameter (910) and the seal to VELO inner diameter (904). This needs to be resolved (action Mike/Trevor)

4.  Exit Window
Mike has posted drawings on EDMS. FEA not yet posted.
Dave reported that Mike confirmed the timetable for the exit window manufacture was compatible with a trial assembly to the GE prior to installation in the cavern.

The beam-pipe meeting is on Tuesday 17 May at CERN (9:00 in room B30/2-005).
5. Beryllium Mirrors
5.1 Prototype: News via Vladimir O, just after meeting: Grinding and polishing in final stages. Now at R=2760, but Vavilov request stopping at 2730 to avoid breaking through glass. This is at limit of specification (1%) - Agreed. We must consider whether subsequent mirror manufacture aims for R=2730 or 2700. Vavilov requested money (<1kCHF) to pay for tool to facilitate measurement of glass thickness (action Carmelo). Delivery date end May was quoted. No news on fixing of bushes (assumed to take place in Vavilov ?)
5.2 Be-mirror Production: No further news regarding ISTC paperwork

5.3 2nd Mirror: order for blank is with Kompozit (but reception not yet confirmed).
5.4 Drawings: No news from Vladimir K. regarding the UIR drawing – needs chasing (action Carmelo/Nick). 
6. Flat Mirrors
RICH2 has received 20 mirrors from the Czech company Compass. Remaining 20 expected end May. So order for RICH1 mirrors should be prepared now (action Nick)
The mirror vertical-size query has been resolved, Richard has updated mirror drawings on EDMS. Need to include roughened area on rear to bond the safety straps (action Richard)
Fabio reported that Christoph had not estimated the natural frequency of the RICH2 mirrors.
Geoff has devised a concept mounting for flat mirrors that allows upper/lower mirrors to be assembled as an 8-mirror plane, but also allows for installation as two separate 4-mirror quadrants. Geoff will meet with Richard next week (18May?) (action Geoff/Richard)

7. Aerogel
Mounting holes for fixation/installation of aerogel required for GE manufacture. Meeting planned for 17th May at CERN (B13/1-017) at 14:00. Aim is to finalize holes and discuss a scheme that avoids operations above/below the beampipe.
8. RICH1 overview meeting

Held at CERN on 3rd May. The planning chart was discussed, and revised following what was agreed at the meeting. The proposed and revised schedules are posted on the CDS agenda. 

10. Next meeting
Friday  27th May at 10:00(GMT) 11:00(CET). 










Dave Websdale

